\begin{thebibliography}{10}

\bibitem{Hatcliff:03}
Hatcliff, J., Deng, W., Dwyer, M., Jung, G., Prasad, V.:
\newblock {Cadena: An Integrated Development, Analysis, and Verification
  Environment for Component-based Systems}.
\newblock In: Proceedings of the 25th International Conference on Software
  Engineering, Portland, OR (May 2003)  160--172

\bibitem{Rajkumar:05}
de~Niz, D., Rajkumar, R.:
\newblock {Partitioning Bin-Packing Algorithms for Distributed Real-time
  Systems}.
\newblock International Journal of Embedded Systems \textbf{2}(3) (2006)
  196--208

\bibitem{Rajkumar:06}
Ghosh, S., Rajkumar, R., Hansen, J., Lehoczky, J.:
\newblock {Integrated QoS-aware Resource Management and Scheduling with
  Multi-resource Constraints}.
\newblock Real-Time Syst. \textbf{33}(1-3) (2006)  7--46

\bibitem{VEST:03}
Stankovic, J.A., Zhu, R., Poornalingam, R., Lu, C., Yu, Z., Humphrey, M.,
  Ellis, B.:
\newblock {VEST: An Aspect-Based Composition Tool for Real-Time Systems}.
\newblock In: RTAS '03: Proceedings of the The 9th IEEE Real-Time and Embedded
  Technology and Applications Symposium, Washington, DC, USA, IEEE Computer
  Society (2003) ~58

\bibitem{Zonghua:04}
Gu, Z., Kodase, S., Wang, S., Shin, K.G.:
\newblock A model-based approach to system-level dependency and real-time
  analysis of embedded software.
\newblock In: RTAS '03: Proceedings of the The 9th IEEE Real-Time and Embedded
  Technology and Applications Symposium, Washington, DC, USA, IEEE Computer
  Society (2003) ~78

\bibitem{GTBook:97}
Rozenberg, G.:
\newblock Handbook of Graph Grammars and Computing by Graph Transformation,
  Volume 1: Foundations.
\newblock World Scientific Publishing Company (jan 1997)

\bibitem{Sendall:03}
Sendall, S., Kozaczynski, W.:
\newblock {Model Transformation: The Heart and Soul of Model-Driven Software
  Development.}
\newblock IEEE Software \textbf{20}(5) (2003)  42--45

\bibitem{Karsai:03}
Karsai, G., Agrawal, A., Shi, F., Sprinkle, J.:
\newblock {On the Use of Graph Transformation in the Formal Specification of
  Model Interpreters}.
\newblock Journal of Universal Computer Science \textbf{9}(11) (2003)
  1296--1321 \url{www.jucs.org/jucs_9_11/on_the_use_of}.

\bibitem{LWCCM}
{Object Management Group}:
\newblock {Lightweight CCM RFP}. {realtime/02-11-27} edn. (November 2002)

\bibitem{Schmidt:08e}
Balasubramanian, K., Schmidt, D.C.:
\newblock {Physical Assembly Mapper: A Model-driven Optimization Tool for
  QoS-enabled Component Middleware}.
\newblock In: Proceedings of the 14th IEEE Real-time and Embedded Technology
  and Applications Symposium, St. Louis, MO, USA (April 2008)  123--134

\bibitem{RTCORBA:05}
{Object Management Group}:
\newblock {Real-time CORBA Specification}. {1.2} edn. (January 2005)

\bibitem{Schmidt:00c}
Wang, N., Schmidt, D.C., Parameswaran, K., Kircher, M.:
\newblock {Applying Reflective Middleware Techniques to Optimize a QoS-enabled
  CORBA Component Model Implementation}.
\newblock In: {24th Computer Software and Applications Conference}, Taipei,
  Taiwan, IEEE (October 2000)

\bibitem{Sharp:98}
Sharp, D.C.:
\newblock {Reducing Avionics Software Cost Through Component Based Product Line
  Development}.
\newblock In: Software Product Lines: Experience and Research Directions.
  Volume 576. (Aug 2000)  353--370

\bibitem{GME:01}
L\'{e}deczi, {\'A}., Bakay, {\'A}., Mar{\'o}ti, M., V{\"o}lgyesi, P.,
  Nordstrom, G., Sprinkle, J., Karsai, G.:
\newblock {Composing Domain-Specific Design Environments}.
\newblock Computer \textbf{34}(11) (2001)  44--51

\bibitem{Subramonian:04c}
Subramonian, V., Shen, L.J., Gill, C., Wang, N.:
\newblock The design and performance of configurable component middleware for
  distributed real-time and embedded systems.
\newblock In: RTSS '04: Proceedings of the 25th IEEE International Real-Time
  Systems Symposium (RTSS'04), Washington, DC, USA, IEEE Computer Society
  (2004)  252--261

\bibitem{Schmidt:05h}
Krishna, A., Gokhale, A., Schmidt, D.C., Hatcliff, J., Ranganath, V.:
\newblock {Context-Specific Middleware Specialization Techniques for Optimizing
  Software Product-line Architectures}.
\newblock In: Proceedings of EuroSys 2006, Leuven, Belgium (April 2006)
  205--218

\bibitem{Hunleth:02}
Hunleth, F., Cytron, R.K.:
\newblock {Footprint and Feature Management Using Aspect-oriented Programming
  Techniques}.
\newblock In: Proceedings of the Joint Conference on Languages, Compilers and
  Tools for Embedded Systems (LCTES 02), ACM Press (2002)  38--45

\bibitem{ByPassIDL:07}
\"{O}mer Erdem~Demir, D\'{e}vanbu, P., Wohlstadter, E., Tai, S.:
\newblock {An Aspect-oriented Approach to Bypassing Middleware Layers}.
\newblock In: AOSD '07: Proceedings of the 6th international conference on
  Aspect-oriented software development, New York, NY, USA, ACM Press (2007)
  25--35

\bibitem{ArnoJacobsenJitM:05}
Zhang, C., Gao, D., Jacobsen, H.A.:
\newblock {Towards Just-in-time Middleware Architectures}.
\newblock In: AOSD '05: Proceedings of the 4th international conference on
  Aspect-oriented software development, New York, NY, USA, ACM Press (2005)
  63--74

\bibitem{BluePencil:06}
Lee, S., Lee, K.W., Ryu, K.D., Choi, J.D., Verma, D.:
\newblock Ise01-4: Deployment time performance optimization of internet
  services.
\newblock Global Telecommunications Conference, 2006. GLOBECOM'06. IEEE (Nov
  2006)  1--6

\bibitem{AutoPerfJ2EE:04}
Diaconescu, A., Mos, A., Murphy, J.:
\newblock Automatic performance management in component based software systems.
\newblock In: ICAC '04: Proceedings of the First International Conference on
  Autonomic Computing (ICAC'04), Washington, DC, USA, IEEE Computer Society
  (2004)  214--221

\bibitem{Quo:97}
Zinky, J.A., Bakken, D.E., Schantz, R.:
\newblock {Architectural Support for Quality of Service for CORBA Objects}.
\newblock Theory and Practice of Object Systems \textbf{3}(1) (1997)  1--20

\bibitem{Schmidt:01s}
Gill, C.D., Cytron, R., Schmidt, D.C.:
\newblock {Middleware Scheduling Optimization Techniques for Distributed
  Real-time and Embedded Systems}.
\newblock In: {Proceedings of the $7^{th}$ Workshop on Object-oriented
  Real-time Dependable Systems}, San Diego, CA, IEEE (January 2002)

\bibitem{zhang:02}
Zhang, R., Lu, C., Abdelzaher, T.F., Stankovic, J.A.:
\newblock {ControlWare: A Middleware Architecture for Feedback Control of
  Software Performance}.
\newblock In: ICDCS '02: Proceedings of the 22 nd International Conference on
  Distributed Computing Systems (ICDCS), Washington, DC, USA (2002)  301

\bibitem{Lu-FCS:02}
Lu, C., Stankovic, J.A., Son, S.H., Tao, G.:
\newblock Feedback control real-time scheduling: Framework, modeling, and
  algorithms.
\newblock Real-Time Syst. \textbf{23}(1-2) (2002)  85--126

\bibitem{Gokhale_GTQMAP_RTAS:08}
Kavimandan, A., Gokhale, A.:
\newblock {Automated Middleware QoS Configuration Techniques using Model
  Transformations}.
\newblock In: Proceedings of the $14^{th}$ IEEE Real-Time and Embedded
  Technology and Applications Symposium (RTAS 2008), St. Louis, MO, USA (April
  2008)  93--102

\bibitem{Deb_EMO:05}
Deb, K., Gupta, H.:
\newblock {Searching for Robust Pareto-Optimal Solutions in Multi-objective
  Optimization}.
\newblock In: Proceedings of the Third International Conference on Evolutionary
  Multi-Criterion Optimization (EMO 2005). Lecture Notes in Computer Science,
  vol. 3410, Springer (March 2005)  150--164

\end{thebibliography}

% $Id: intro.tex 16889 2009-02-21 03:51:46Z gokhale $
\section{Introduction}
\label{sec:intro}
Component-based software engineering (CBSE) has received much attention
over the past few years to develop distributed systems including
distributed, real-time, and embedded (DRE) systems, such as emergency
response systems, aircraft navigation and command supervisory systems,
and total shipboard computing systems.  CBSE provides a simplified
programming model and various mechanisms to separate functional and
non-functional concerns of the system being designed, which lends it to
rapid prototyping, (re-) configuration, and easier maintenance of DRE
systems.

DRE systems have stringent runtime quality of service (QoS) requirements
including predictable end-to-end latencies, reliability and security,
among others.  Naturally, the software architecture of the DRE system
plays an important role in ensuring that the runtime QoS needs of DRE
systems are met.  In component-based DRE systems, the software
architecture is defined not just by the source code of the application
functionality but also by a wide range of systemic issues including the
assembly of application components, their deployment on the target nodes
of the system and allocation of resources such as the CPU, and the
component middleware that hosts the application components. 

In component middleware, such as the CORBA Component Model (CCM) and
Enterprise Java Beans (EJB), a container is a central concept that hosts
application components.  Containers hosting DRE system components 
provide a high degree of configurability by allowing (1) the choice of
the number of thread resources to be configured for each component, 
their type (\emph{i.e.}, static or dynamic), and their attributes, such
as stacksize, etc., (2) control over asynchronous event communication,
and event filtering and delivery options, and (3) control over client
request invocation priorities on the server component.   The
\emph{configuration space} -- identified by all the mechanisms for
specifying system QoS and their appropriate values -- becomes highly 
complex.  Thus, making the right configuration decisions is one key
factor that determines the quality of the DRE system software
architecture.

Prior research in improving the quality of DRE systems software
architectures has focused on: (1) analysis-driven
decomposition~\cite{Hatcliff:03} of DRE system functionality into
reusable application components that can be assembled and deployed; (2)
component-to-node assignment~\cite{Rajkumar:05} and resource
allocations~\cite{Rajkumar:06}, and (3) schedulability and timing
analysis~\cite{VEST:03,Zonghua:04} to determine exact priorities and
time periods for applications, which help in partitioning the system
resources and configuring the middleware.

Despite these advances, key issues still remain unresolved in the
deployment and configuration problem space of DRE system software
architectures.  For example, although bin packing
algorithms~\cite{Rajkumar:05} make effective decisions on component 
deployment, and schedulability analysis determines the priorities at
which the components should execute, both these decisions are at best
coarse-grained since they determine only the nodes on which the
components must be deployed but do not indicate how they are deployed
within the containers of the component middleware.  The lack of
finer-grained decisions often lead to suboptimal runtime QoS since these
decisions are now left to application developers, who are domain experts
but often lack detail understanding of the middleware.

To address these limitations, this paper presents a heuristics-based
algorithm implemented within a
model-transformation~\cite{GTBook:97,Sendall:03}  framework that
combines models of user-specified QoS requirements, node assignment
decisions, and priority values, and transforms them into optimal 
middleware configurations thereby enhancing the quality of the DRE
system software architecture.  Our research prototype has been
implemented using the GReAT~\cite{Karsai:03} model transformation
framework for the Lightweight CCM (LwCCM)~\cite{LWCCM} middleware.

Two significant benefits accrue from our approach.  First, by realizing
the heuristic-based algorithm as an automated model transformation
process, it can be seamlessly reapplied and reused during the iterative
DRE system development process.  Second, the configurations generated by
our algorithm can be leveraged by additional backend optimization tools
and techniques, such as the Physical Assembly Mapper
(PAM)~\cite{Schmidt:08e} which reduces time and space overheads by
merging collocated components at system deployment-time. 

This paper is organized as follows: Section~\ref{sec:chal} discusses the
challenges developers face in achieving optimal QoS configuration for
DRE systems;  Section~\ref{sec:sol} presents the overall approach taken,
the enabling technologies  used in our technique, and the model
transformation algorithm  we have developed; Section~\ref{sec:results}
empirically evaluates our approach in the context of a representative
case study;  Section~\ref{sec:related}  discusses the related work in
the area;  Section~\ref{sec:conclusion} gives concluding remarks.


%%% Local Variables: 
%%% mode: latex
%%% TeX-master: "optimization"
%%% End: 
